Author's response to reviews

Title: The Association of Sedentary Behaviors and Body Mass Index Classification in a Cross-Sectional Analysis: Are the Effects Homogenous?

Authors:

Justin B Dickerson (dickerson@srph.tamhsc.edu)
Matthew Lee Smith (matthew.smith@srph.tamhsc.edu)
Mark E Benden (mbenden@srph.tamhsc.edu)
Marcia G Ory (mory@srph.tamhsc.edu)

Version: 4 Date: 13 November 2011

Author's response to reviews: see over
Dear Editor Lock,

Thank you for the kind and helpful feedback regarding our manuscript. We have attempted to address the concerns of the Reviewers in italics below, and hope that you will find our changes and explanations sufficient to warrant publication in *BMC Public Health*. Best regards.

______________
Dr. Nanchahal’s Comments to Authors:

*We are thankful for Dr. Nanchahal’s very helpful comments, and glad those comments have now been addressed to Dr. Nanchahal’s full satisfaction.*

______________
Dr. Buchowski’s Comments to Authors:

*The conclusion in abstract does not match conclusions in the manuscript. Statement: "The sedentary behavior of watching TV television exhibited a differential relationship with BMI depending on BMI classification" is somewhat awkward.*

*The authors have deleted this sentence from the abstract to eliminate this confusion.*

______________
Needs some language corrections before being published.

*As is customary, the authors are more than willing to work with the journal’s copy-editing team to address any issues in the publication process.*